Author’s response to reviews

Title: Management Of Arrhythmogenic Right Ventricular Dysplasia In Pregnancy: A Rare Case

Authors:

Nilgün Güdücü (nilgun.kutay@gmail.com)
Salih S Kutay (serdarkutay@hotmail.com)
Ebru Özenç (eozenc@hotmail.com)
Çavlan Çiftçi (cavlanciftci@superonline.com)
Alin B Yi (alinbasgul@gmail.com)
Herman (hermani@superonline.com)

Version: 2 Date: 20 November 2010

Author’s response to reviews:

Dear Editor,
We have made the suggested structural changes as:
The title mistake revised as Dysplasia
Abstract is changed as to include sections introduction, case presentation, conclusion
Abstract changed as 'an autosomal dominantly inherited'
Title page added as page 1
Consent section included
Authors contributions included
Best Regards,
Nilgün Güdücü